Author's response to reviews

Title: Leveling off of prevalence of obesity in the adult population of Sweden between 2000/01 and 2004/05

Authors:

Jan Sundquist (jan.sundquist@med.lu.se)
Sven-Erik Johansson (sven-erik.johansson@med.lu.se)
Kristina Sundquist (kristina.sundquist@ki.se)

Version: 6 Date: 6 March 2010

Author's response to reviews:

All the authors have read the final version of the manuscript (all previous versions too)

We have changed the webadress on page 10:

For example, during these four years the Swedish National Institute of Public Health (http://www.fhi.se/en/) has emphasized increased physical activity and better eating habits as two of the most significant areas to focus on in order to improve public health in Sweden.

Best wishes,

Jan Sundquist